Author’s response to reviews

Title: Prevalence and Causes of Blindness, Visual Impairment Among Different Ethnical Minority Groups In Xinjiang Uygur Autonomous Region, China

Authors:

Yanping Li (5861623@qq.com)
Wenyong Huang (1220792380@qq.com)
Qoyun Qiqige (aoyunqqg@163.com)
Hongwei Zhang (zhwzhw.123@163.com)
Ling Jin (lingjin41@yahoo.com)
Pula Ti (anrabc@126.com)
Jennifer Yip (jennifer.yip@lshtm.ac.uk)
Baixiang Xiao (xiaobaixiang2006@126.com)

Version: 4 Date: 28 Jan 2018

Author’s response to reviews:

Thanks for the email on 25th Jan.

The following changes have been made according to your suggestions:

1. Ethical approval of the content and the way of verbal consent to the study sought by the field team leader before eye examinations.

2. The figure legend to explain that the picture was taken from wiki website.

3. The data link is working and had been tested. Please check and may try again if needed.